Author's response to reviews

Title: Cystic colon duplication causing intussusception in a 25-year-old man: report of a case and review of the literature

Authors:

Carolin Reiser-Erkan (carolin.reiser@gmx.de)
Mert Erkan (mert.erkan@gmx.de)
Erika Ulbrich (erika.u@gmx.de)
Jörg Nährig (Joerg.Naehrig@lrz.tu-muenchen.de)
Jörg Kleeff (kleeff@gmx.de)

Version: 5 Date: 4 June 2010

Author's response to reviews: see over
MS: 1091012756302452 “Cystic colon duplication causing intussusception in a 25-year-old man: report of a case and review of the literature”
Carolin Reiser-Erkan, Mert Erkan, Erika Ulbrich, Jörg Nährig and Jörg Kleeff

Point-by point-response – June 04, 2010

Editorial points:

Major revisions:

1. Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

   Answer:
   We wrote each affiliation in the required format.

2. Figures: Please provide figure titles. These should be concise and be the first sentence of the figure legend, preferably indicated with bold text. These are titles for the whole figure, not the labels for A, B and C which you have already provided.

   Answer:
   We provided figure titles for both figures with bold text.

3. Manuscript: this manuscript does not have the correct structure.

   Please reformat it to use these headings: Abstract; Background; Case Presentation; Conclusions; Abbreviations (if any); Competing interests; Authors' contributions; Authors’ information (optional); Acknowledgements (optional); References.

   Answer:
   We reformatted the manuscript to the correct structure. There are no abbreviations in the manuscript, so we left this heading out.
We removed the running title like requested.

We removed the authors’ qualifications from the manuscripts as well as from the submission system.

We removed the word count, key words and the text at the bottom of the front page: “received: revised: accepted: published online:”.

We moved the figure legends to after references.

We checked the manuscript for any typographical errors and made final corrections marked with red.